Author’s response to reviews

Title: Tigecycline-induced sustained and severe hypoglycemia: a case report

Authors:

Yixin Chen (yixin29@zju.edu.cn)
Lin Li (3312012@zju.edu.cn)
Nan Zhang (3198047@zju.edu.cn)
Hong Li (srrshnfm@zju.edu.cn)

Version: 1 Date: 20 Mar 2019

Author’s response to reviews:

Dear, Editors:

On behalf of my co-authors, I am submitting the revised material “Tigecycline-induced sustaining and severe hypoglycemia: a case report” for possible publication in BMC Pharmacology and Toxicology.

This manuscript has been extensively edited by a native English professional with science background at Elixigen Corporation. We certify that we have participated sufficiently in the work to take public responsibility for the appropriateness of the collection, analysis, and interpretation of the data.

We have reviewed the final version of the manuscript and approved it for publication. To the best of our knowledge and belief, this manuscript has not been published in whole or in part nor is it being considered for publication elsewhere.

Best Regards.

Yours sincerely,
Hong, Li
Institute: Zhejiang University School of Medicine Sir Run Run Shaw Hospital, Endocrinology
Address: east Qing chun road No.3, Hangzhou, China, CN 310016,
Email: srrshnfm@zju.edu.cn
Tel: 13588706366